Author's response to reviews

Title: Nurses' preparedness to care for women exposed to Intimate partner violence: a quantitative study in primary health care.

Authors:

Eva M Sundborg (eva.sundborg@sll.se)
Nouha Saleh Stattin (nouha.saleh-stattin@sll.se)
Per Wändell (per.wandell@ki.se)
Lena Törnkvist (lena.tornkvist@sll.se)

Version: 8 Date: 9 January 2012

Author's response to reviews:

Figures: The figure is now cited in the manuscript.
Figure 1 has been corrected